Author’s response to reviews

Title: Primary extracranial meningioma in submandibular region - A case report

Authors:

Sanjay D Deshmukh (drsanjay123in@yahoo.co.in)
Vidya V Rokade (vidya_rokade@rediffmail.com)
Gayatri S Pathak (drgayatridp@yahoo.in)
Sanjana V Nemade (drsanjana31@yahoo.in)
Amrut V Ashturkar (amrutva@yahoo.com)

Version: 3 Date: 10 January 2011

Author’s response to reviews:

To, Respected Editor, Attaching the revised manuscript for your kind perusal. Regrets for sending it late due to some unavoidable circumstances. Thanking you, Dr S D Deshmukh.

1. Additional references have been incorporated as suggested.
2. Consistency has been maintained with regards to the differential diagnosis.
3. Inadequacies regarding the authorship have been addressed.

The changes have been highlighted in red colour.